Author’s response to reviews

Title: Thailand Medical Mobile Application for Patients Triage Base on Criteria Based Dispatch Protocol

Authors:

Krongkarn Sutham (krongkarn.s@cmu.ac.th)
Pattaraporn Khuwuthyakorn (Pattaraporn.Khuwuthyakorn@cmu.ac.th)
orawit Thinnukool (orawit.t@cmu.ac.th)

Version: 2 Date: 06 Mar 2020

Author’s response to reviews:

COVER LETTER

Dear Editor

I am enclosing here with a manuscript entitled “Thailand Medical Mobile Application for Patients Triage Base on Criteria Based Dispatch Protocol submitted to “BMC journal” for revision version evaluation round 2.

The answer of revision version are;

1) The English is relatively poor and needs to improve quality
Fixing Expatiation: The proof reading has been conducted by native speaker.

2) I do not think you mean to use the word "pain" and may have meant to use "main"
Fixing Expatiation: The sentence has been fixed to another sentence;

“In summary, the Thailand mobile medical application for patient triage is smart technology, which can benefit users as a tool for rapid triage screening, educate patients and new emergency staff, and prevent the overcrowding of emergency care units in Thai hospitals”

Corresponding Author’s Name
Assistant Professor Orawit Thinnukool, Ph.D.